Homework1

Submission

Submit your individual so lut ion at the beginning ofthe class on Monday, September 11.

Proble ms

1. Prove by induction that the number ofvertices in a binar y tree ofheight n is at most 1

2 1 - + n .

2. Problem18 on page 14.

3. Problem2 on page 26.

4. Show that for languages

3 2 1

, , L L L , if ) (

3 2 1

L L L w Ç Î , then

3 1 2 1

L L L L w Ç Î .

Find 3 2 1 , , L L L such that there is a 3 1 2 1 : LL L L w w Ç Î and ) ( 3 2 1 L L L w Ç Ï .

5. Problem2 on page 47.

Good luc k!